Author's response to reviews

Title: Methods for identifying surgical wound infection after discharge from hospital: a systematic review

Authors:
Emily S Petherick (ep9@york.ac.uk)
Jane E Dalton (jd518@york.ac.uk)
Peter J Moore (enid.bridge@nlq.nhs.uk)
Nicky Cullum (nac2@york.ac.uk)

Version: 8 Date: 16 November 2006

Author's response to reviews: see over
Thursday, 16 November 2006

To the Editor

MS 1223656836934427
Methods for identifying surgical wound infection after discharge from hospital: a systematic review

Thank you for the news of our manuscript acceptance in BMC Infectious diseases. We have made the following changes to the manuscript as requested. The changes made were;

1) Title of the manuscript has been changed to sentence case
2) Full stop has been removed from the manuscript title
3) Tables have been checked and are now numbered consecutively within the text
4) Text relating to placement of tables within the text has been removed
5) Table 2 (now table 3) has been amended so that it is now one table

We trust that these changes are satisfactory and we look forward to publishing with you again.

Kind Regards

Emily Petherick, on behalf of all study authors.